\documentclass[11pt]{article}
\usepackage{graphicx}
\usepackage{amssymb}
\usepackage{epstopdf}
\DeclareGraphicsRule{.tif}{png}{.png}{`convert #1 `dirname #1`/`basename #1 .tif`.png}

\textwidth = 6.5 in
\textheight = 9 in
\oddsidemargin = 0.0 in
\evensidemargin = 0.0 in
\topmargin = 0.0 in
\headheight = 0.0 in
\headsep = 0.0 in
\parskip = 0.2in
\parindent = 0.0in

\newtheorem{theorem}{Theorem}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{definition}{Definition}

\begin{document}



\begin{center}
{\bf Math 226.01: Differential Equations and Linear Algebra }
\\
{\bf Quiz 5}
\\
{\bf October 8, 1998}
\end{center}
\vspace*{.4cm}

\underline{{\bf Name:}\hspace*{10cm}}


Find a particular solution of the following inhomogeneous differential equation
\[
    y''- 4y'+5y=(17 t-8)e^{-2t}.
\]


\vfill
\underline{{\bf Answer:}\hspace*{10cm}}

Sign the pledge:
``On my honor, I have neither given nor received unauthorized aid on this Exam.''

\underline{{\bf Signature:}\mbox{\hspace*{10cm}}}



 \end{document} 